Author's response to reviews

Title: Study on the clinical application of pulsed DC magnetic technology for tracking of intraoperative head motion during frameless stereotaxy

Authors:

Olaf Suess (olaf.suess@charite.de)
Silke Suess (silke.suess@charite.de)
Sven Mularski (sven.mularski@charite.de)
Bjorn Kuhn (bjoern.kuehn@charite.de)
Thomas Picht (thomas.picht@charite.de)
Stefanie Hammersen (stefanie.hammersen@charite.de)
Rudiger Stendel (ruediger.stendel@charite.de)
Mario Brock (mario.brock@charite.de)
Theodoros Kombos (theodoros.kombos@charite.de)

Version: 4 Date: 23 April 2006

Author's response to reviews:

Dear Editors:

please find enclosed my final version of the above mentioned paper with the following formatting changes:
1) all unnecessary capitilisation is removed from the titel
2) the competing interest declaration is changed to "... no competing interests."
3) the references are renumbered, starting with Ref#1.

Thank you once again for publishing my manuscript.
With kindest regards,
Dr. med. O.Suess